\begin{table}%t1
\caption{\label{tab1}The list of $\lambda$ Boo stars.  }
%\centering
\par
\begin{tabular}{rccccccccccc}
\hline\hline\noalign{\smallskip}
       HD~&  G & P1 & AM & H & A & L & AJ & V  & P2 &  Spectral classification  & Remarks \\  
\noalign{\smallskip}\hline   \noalign{\smallskip}                     
        3 &    &    & AM &   &   &   &    &   &    &                           & inconsistent UV flux (bin.?)         \\
      319 &  G & P1 &    & H & A &   &    & V  & P2 & A1mA2 Vb $\lambda$Boo PHL &        \\
     2904 &    &    & AM &   &   &   &    &   &    &                           &  inconsistent UV flux (bin.?)       \\                                 
     4158 &    & P1 &    & H &   &   &    &   & P2?&                           &        \\        
     5789 &    &    & AM &   &   &   &    &   &    &                           &         \\     
     6173 &    &    &    &   & A &   &    &   &    &                           &          \\                   
     6870 &    & P1 &    &   &   &   &    &   & P2 &                           & misclassified: \\            
     7908 &    &    &    &   &   &   &    &   & P2 &                           &              \\              
    11~413 &  G & P1 & AM & H & A &   &    & V  & P2 &   A1 Va $\lambda$Boo PHL  &               \\
    11~503 &    &    & AM &   &   &   &    &   &    &        B9.5 IV$^{+}$n     &              \\
    13~755 &    &    &    &   &   &   &    &   & P2 &                           &              \\              
    15~165 &    &    &    &   &   &   &    &   & P2 &                           &                \\                
    16~955 &    &    &    & H & A &   &    &   &    &                           &               \\         
    21~335 &    &    &    & H & A &   & AJ &   &    &                 A3 IVn    &               \\
    22~470 &    &    & AM &   &   &   &    &   &    &                           & composite, Hipparcos\\   
    23~258 &    &    & AM &   &   &   &    &   &    &                           &                   \\  
    23~392 &  G &    &    &   &   &   &    &   & P2 & A0 Va$^{-}$ ($\lambda$Boo) NHL &  inconsistent UV flux (bin.?) \\
    24~472 &    &    &    &   &   &   &    &   & P2 &                           &  inconsistent UV flux (bin.?)               \\                
    26~801 &    &    &    &   & A &   &    &   &    &                           &               \\
    30~422 &  G & P1 &    &   &   &   &    & V  & P2 &    A3 Vb $\lambda$Boo PHL &           \\
    30~739 &    &    & AM &   &   &   &    &   &    &         A0.5 IVn          &  inconsistent UV flux (solar ab.)              \\
    31~295 &  G & P1 & AM & H &   &   & AJ & V  & P2 &    A0 Va $\lambda$Boo NHL &              \\
    34~787 &    &    & AM &   &   &   &    &   &    &                           & misclassified     \\             
    35~242 &    &    &    &   &   &   &    & V  & P2 &                           &     \\        
    36~496 &    &    &    &   &   &   & AJ &   &    &                           & composite, Hipparcos      \\             
    36~726 &  G &    &    &   &   &   &    &   & P2 & kA0hA5mA0 $\lambda$BooNHL &       \\
    37~411 &  G &    &    &   &   &   &    &   &    & kA0hA3mA0Va(e)/shell $\lambda$Boo &       \\
    37~886 &    &    &    &   &   & L &    &   &    &                           & misclassified     \\                               
    38~545 &  G & P1 &    &   &   &   & AJ &   & P* & A2 Va$^{+}$ $\lambda$Boo PHL &  composite, Hipparcos      \\  
    39~283 &    &    &    &   &   &   & AJ &   &    &           A1 Va       &   inconsistent UV flux (solar ab.)      \\ 
    39~421 &    & P1 &    & H & A &   &    & V  & P* &        A1 Van            &            \\
    41~580 &    &    &    &   & A &   &    &   &    &                          &         \\
    47~152 &    &    & AM &   &   &   &    &   &    &                          & composite, Hipparcos        \\ 
    54~272 &    &    &    &   &   &   &    &   & P2 &                          &  inconsistent UV flux (bin.?)    \\  
    56~405 &    &    &    &   & A &   &    & V  &    &          A2 Va           &  inconsistent UV flux (bin.?)         \\
    64~491 &  G &    & AM &   &   &   &    &   & P2 &    kA3hF0mA3 V  $\lambda$Boo (PHL) & composite, spectrum    \\
    66~684 &    &    & AM &   &   &   &    &   &    &          B9 Va           &  discordant $v\sin i$ values    \\
    74~873 &  G &    &    &   &   &   &    &   & P2 &     kA0.5hA5mA0.5 V  $\lambda$Boo NHL &  inconsistent UV flux (solar ab.)  \\  
    75~654 &    & P1 &    &   &   &   &    & V  & P2 &                         &        \\      
    79~108 &    &    & AM &   &   &   &    & V  &    &                      &  $RV$ variable        \\           
    81~104 &    &    &    &   & A &   &    &   &    &                          & composite, spectrum         \\
    81~290 &    & P1 &    &   &   &   &    &   & P2 &                          &     \\    
    83~041 &    & P1 &    &   &   &   &    &   & P2 &      FHB F2VkA3mA2 \lambo~PHL & misclassified   \\
    83~277 &    &    &    &   &   &   &    &   & P2 &                          &   inconsistent UV flux (bin?)        \\         
    84~123 &    & P1 &    &   &   &   &    &   & P2 &                         & misclassified:      \\            
    84~948 &    & P1 &    &   &   &   & AJ &   & P2 &                        &  SB2, metal ab. to be reviewed       \\        
    87~271 &  G &    &    &   &   &   &    &   &    &          A9kA0mA0 V  $\lambda$Boo  &     \\
    87~696 &    &    & AM &   &   &   &    &   &    &                     A7 V    &  inconsistent UV flux (solar ab.)         \\
    89~239 &    &    & AM &   &   &   &    &   &    &                           &    inconsistent UV flux (solar ab.)      \\
    89~353 &    &    &    &   & A &   &    &   &    &                           & misclassified        \\                        
    90~821 &  G &    &    &   &   &   &    &   &    &           kA2hA7mA2 Vn $\lambda$Boo &        \\
    91~130 &  G &    & AM &   &   &   &    &   &    &            A0 Va$^{-}$ $\lambda$Boo (PHL) & \\
    97~773 &    &    &    &   &   &   & AJ? &  &    &                           & composite, Hipparcos                \\              
    97~937 &    &    &    &   & A &   &    &   &    &                           &                 \\            
    98~353 &    &    &    & H & A &   & AJ &   &    &  A1 Va (composite spectrum ?) & composite, spectrum        \\
    98~772 &    & P1 & AM &   &   &   &    &   & P* &                           &             \\            
   101~108 &    & P1 &    & H &   &   & AJ &   & P* &                           &             \\   % 
102~541 &    &    &    &   &   &   &    &   & P2 &                           &  inconsistent UV flux (bin.?)             \\       
   105~058 &  G & P1 &    & H &   &   & AJ &   & P2 &    kA1hA7mA1 V   $\lambda$Boo (PHL) &      \\
   105~759 &  G &    &    &   &   &   &    &   &    &             kA2hF0mA2 V  ($\lambda$Boo)  &   \\
   106~223 &    & P1 &    & H &   &   & AJ &   & P2 &         F3 V  m-3    &  misclassified       \\                    
   107~233 &  G & P1 &    &   &   &   &    &   & P2 & kA1hF0mA1 Va $\lambda$Boo PHL  &      \\
   108~283 &    &    & AM &   &   &   &    &   &    &                A9 IVnp SrII   & misclassified       \\ 
   109~738 &    & P1 &    &   &   &   &    &   & P2 &                           &            \\                      
   109~980 &    &    & AM &   &   &   &    &   &    &                           &   \\    
   110~377 &    &    & AM &   &   &   &    &   & P2 &                           &      \\                      
   110~411 &  G & P1 &    &   &   &   & AJ &   & P2 &     A0 Va ($\lambda$Boo) NHL &      \\
   111~005 &    &    &    &   &   &   &    &   & P2 &                           & D, Hipparcos       \\                   
   111~604 &    &    & AM &   &   &   &    &   & P2 &                           & $RV$ variable    \\                  
   111~786 &  G & P1 & AM & H &   &   &    &   & P2 &      A1.5 Va$^{-}$ $\lambda$Boo PHL & composite, spectrum  \\  
   112~097 &    &    & AM &   &   &   &    &   &    &                           &            \\ 
   118~623 &    &    & AM &   &   &   &    &   &    &         A7 Vn             & composite, Hipparcos    \\  
   120~500 &  G &    &    &   &   &   &    &   & P2 &      kA1.5hA5mA1.5 V  ($\lambda$Boo) NHL &  \\  
   120~896 &    &    &    &   &   &   &    &   & P2 &                           &    \\                    
   125~162 &  G & P1 & AM & H &   &   & AJ &   & P2 &  A0 Va $\lambda$Boo NHL   &       \\
   125~489 &    &    & AM &   &   &   &    &   &    &           A7 Vn           &  inconsistent UV flux (bin.?)       \\ 
   125~889 &    &    &    &   &   &   &    &   & P2 &                           &    \\                  
   130~158 &    &    & AM &   &   &   &    &   &    &                           &     \\  
   130~767 &    &    &    &   &   &   &    &   & P2 &                           &      \\                      
   138~527 &    &    & AM &   &   &   &    &   &    &                           &  D, Hipparcos  \\ 
   141~851 &    & P1 &    & H & A &   &    &   & P* &                           & composite, AO     \\          
   142~703 &  G & P1 & AM &   &   &   &    &   & P2 &        kA1hF0mA1 Va $\lambda$Boo PHL & \\
   142~994 &  G & P1 &    &   &   &   &    &   & P2 &     A3 Va $\lambda$Boo PHL &         \\
   144~708 &    &    & AM &   &   &   &    &   &    &                           &     \\ 
   148~638 &    &    &    &   &   &   &    &   & P2 &                           &  D, Hipparcos   \\        
   149~130 &    &    &    &   &   &   &    &   & P2 &                           &     \\      
   149~303 &    & P1 &    &   & A &   &    &   & P* &                           &      \\  
   153~747 &    &    &    &   &   &   &    &   & P2 &                           &      \\                
   153~808 &    &    & AM &   &   &   &    &   &    &              A0 IV$^{+}$  & composite, spectrum       \\
   154~153 &    &    &    &   &   &   &    &   & P2 &      F1 V  m-2.5           & misclassified      \\
   156~954 &    & P1 &    &   &   &   &    &   & P2 &                           &  inconsistent UV flux (bin.?)    \\            
   159~082 &    &    & AM &   &   &   &    &   &    &                           &   inconsistent UV flux (bin.?)    \\ 
   160~928 &    & P1 &    &   &   &   &    &   & P* &                           & composite, interferometry     \\     
   168~740 &    & P1 &    &   &   &   &    &   & P2 &                           &     \\                   
   168~947 &    &    &    &   &   &   &    &   & P2 &                           &  inconsistent UV flux (bin.?)    \\                  
   169~009 &    &    & AM &   &   &   &    &   &    &                           &  $RV$ variable    \\ 
   169~022 &    &    &    & H &   &   &    &   &    &    A0II$^-$(n)shell       &misclassified      \\
   170~000 &    &    & AM &   &   &   &    &   &    &      kB9hB9HeA0V(Si)      & composite, Hipparcos      \\
   170~680 &  G & P1 & AM &   &   &   &    &   & P2 &    A0 Van ($\lambda$Boo) NHL &  inconsistent UV flux (solar ab.)   \\
   171~948 &  G & P1 &    &   & A &   &    &   & P2 &     A0 Vb $\lambda$Boo NHL & SB2, twin stars?     \\
   174~005 &  G &    &    &   &   &   &    &   &    &      A7V  kA2mA2 $\lambda$Boo & composite, spectrum    \\
   175~445 &    &    &    &   &   &   &    &   & P2 &                           &  inconsistent UV flux (bin.?)          \\                     
   177~120 &    & P1 &    &   & A &   &    &   & P* &                           &   inconsistent UV flux (bin.?)    \\      
   177~756 &    &    & AM &   &   &   &    &   &    &       kB8HeA0IV wk4481    & $RV$ variable      \\
   179~791 &    &    &    & H & A &   &    &   &    &                           &  inconsistent UV flux (solar ab.)     \\
   183~324 &  G & P1 & AM & H & A &   & AJ &   & P2 &     A0 Vb $\lambda$Boo NHL&  $RV$ variable     \\
   184~190 &    & P1 &    &   &   &   &    &   &    &                           &        \\
   184~779 &    & P1 &    &   &   &   &    &   & P2 &                           &      \\                   
   191~850 &    &    &    &   &   &   &    &   & P2 &                           &      \\                   
   192~424 &    & P1 &    &   & A &   &    &   &    &                           &       \\ 
   192~640 &  G & P1 & AM & H &   &   &    &   & P2 &   A0.5 Va$^-$ $\lambda$Boo PHL & discordant $v\sin i$ values    \\
   193~063 &    &    &    &   & A &   &    &   &    &                           &          \\     \hline 
   193~256 &  G & P1 &    &   &   &   &    &   & P2 &    A2 Va $\lambda$Boo PHL  &  D, Hipparcos         \\
   193~281 &  G & P1 &    &   &   &   &    &   & P2 &    A3mA2 Vb $\lambda$Boo PHL  &      \\
   196~821 &    &    & AM &   &   &   &    &   &    &                            &   inconsistent UV flux (bin.)       \\ 
   198~160 &  G & P1 &    &   &   &   &    &   & P2 &    A2 Vann $\lambda$Boo PHL: &  metal ab. to be reviewed     \\
   198~161 &  G &    &    &   &   &   &    &   &    &                     A2    &  metal ab. to be reviewed       \\  
   200~841 &    &    &    &   & A &   &    &   &    &                           &   inconsistent UV flux (bin.?)    \\                        
   204~041 &  G & P1 & AM &   &   &   &    &   & P2 &   A1 Vb $\lambda$Boo PHL  &   inconsistent UV flux (bin.? )     \\
   204~965 &    &    &    &   & A &   &    &   &    &                           &        \\ 
   210~111 &  G & P1 &    &   &   &   &    &   & P2 &   kA2hA7mA2 Vas $\lambda$Boo PHL &  \\
   210~418 &    &    &    & H & A &   &    &   &    &              A2mA1 IV-V (SB2)   & composite, spectrum \\
   212~150 &    &    & AM &   &   &   &    &   &    &                             &  inconsistent UV flux (bin.?)    \\ 
   214~454 &    &    & AM &   &   &   &    &   &    &               A9 V  kA7mA6   &    \\
   216~847 &    &    &    &   &   &   &    &   & P2 &                             & inconsistent UV flux (bin.?)      \\                  
   217~782 &    &    &    &   &   &   & AJ &   &    &                             & composite, Hipparcos \\  
   218~396 &  G &    &    &   &   &   &    &   &    &       kA5hF0mA5 v \lambo~   & \\
   220~061 &    &    & AM & H & A &   &    &   &    &       A8 V  kA5mA5 $\lambda$4481weak & $RV$ variable \\
   220~278 &    &    &    & H & A &   &    &   &    &               A5 Vn         & composite, Hipparcos \\
   221~756 &  G & P1 &    &   & A &   & AJ &   & P2 &      A1 Va$^{+}$ ($\lambda$Boo) P/NHL & inconsistent UV flux (solar ab.)  \\
   223~352 &    &    & AM &   &   &   &    &   &    &        A0Va$^{+}$n         &            \\
   225~218 &    &    & AM &   &   &   &    &   &    &                            & inconsistent UV flux (bin.)    \\ 
   261~904 &    &    &    &   &   &   &    &   & P2 &                            &      \\    
   290~492 & G  &    &    &   &   &   &    &   & P2 &  A0.5 Vb ($\lambda$Boo)   NHL & composite, interferometry \\
   290~799 &  G & P1 &    &   &   & L &    &   & P2 &  A2 Vb $\lambda$Boo PHL    &        \\
   294~253 &  G & P1 &    &   &   & L &    &   & P2 &     B9.5 Va ($\lambda$Boo) NHL & \\
\hline
\end{tabular}
\smallskip
The spectral classification is taken  from Gray and co-authored papers.\\
Symbols of the columns: \\
G: Gray (his website) and stars added in published papers (see text).     \\
P1: Paunzen et~al. (\cite{Paunzen1997e}).    \\
AM: Abt \& Morrell (\cite{Abt1995}).  \\
H:  Hauck  \& Sletteback (\cite{Hauck1983}),  Hauck (\cite{Hauck1986}).\\
A: Abt (\cite{Abt1984a}, \cite{Abt1984b}, \cite{Abt1985}).     \\
L: Levato et~al. (\cite{Levato1994}).    \\
AJ: Andrillat et~al. (\cite{Andrillat1995}).   \\
V: Vogt et~al. (\cite{Vogt1998}).     \\
P2: Paunzen (\cite{Paunzen2001}).    \\
P*: classified as non-\lambo~in Paunzen et~al. (\cite{Paunzen2001e}),  see text. 
\end{table}